Author's response to reviews

Title: Outcomes of Total Hip Arthroplasty in Patients with Primary Immune Thrombocytopenia

Authors:

Seung-Jae Lim (limsj70@gmail.com)
In-Kwon Yeo (ikyeo@samsung.com)
Chan-Woo Park (cwpark@samsung.com)
Young-Wan Moon (ywmoon@skku.edu)
Youn-Soo Park (ysp3504@skku.edu)

Version: 2
Date: 24 January 2015

Author's response to reviews:

We have revised our manuscript according to the editorial request.

1. Line Numbering
   # We have added a line numbering.

2. Ethics Statement
   # We have described an Ethics Statement in the text.

3. Consent Statement
   # We have described a Consent Statement in the text.

4. List of Abbreviation
   # n/a

5. Acknowledgements
   # We have described an Acknowledgement in the text.